Author’s response to reviews

Title: MOG antibody seropositivity in a patient with encephalitis: beyond the classical syndrome

Authors:

Sara Mariotto (sara.mariotto@gmail.com)
Salvatore Monaco (salvatore.monaco@univr.it)
Patrick Peschl (patrick.peschl@i-med.ac.at)
Ilaria Coledan (ilaria.coledan@alice.it)
Romualdo Mazzi (romualdo.mazzi@aovr.veneto.it)
Romana Höftberger (omana.hoeftberger@meduniwien.ac.at)
Markus Reindl (markus.reindl@i-med.ac.at)
Sergio Ferrari (sergio.ferrari@aovr.veneto.it)

Version: 2 Date: 12 Sep 2017

Author’s response to reviews:

Dear Editor,

Thanks for your comments.

We now clarified that a written consent was obtained from the patient and we changed the form accordingly.

We also removed the CARE Checklist and uploaded a clean copy of the manuscript.

Thank you for your consideration of this manuscript.

Sincerely,

Sara Mariotto

Department of Neurosciences, Biomedicine, and Movement Sciences, University of Verona, Italy